Author’s response to reviews

Title: Gender-based violence and engagement in biomedical HIV prevention, care and treatment: A scoping review

Authors:

Anna Leddy (annamleddy@gmail.com)
Ellen Weiss (eweiss@popcouncil.org)
Eileen Yam (eyam@popcouncil.org)
Julie Pulerwitz (jpulerwitz@popcouncil.org)

Version: 2 Date: 03 May 2019

Author’s response to reviews:

Associate Editor Comments

Thank you for addressing the comments from the initial reviewers. I find that the paper is acceptable for publication. There was one reference Roberts (2016) that does not correspond to the reference list (which cites 1999). As you know, PrEP was not available in 1999. This must be reconciled before publication, but I am accepting the paper because it is a minor detail that I believe can be easily rectified.

- We have updated the citation so it now indicates the correct date of publication. The revised citation (#75) can be found at the bottom of page 26. It now reads: